Author's response to reviews

Title: Television viewing in Thai infants and toddlers: impacts to language development and parental perceptions

Authors:

   Nichara Ruangdaraganon (kositprapa@hotmail.com)
   Jariya Chuthapisith (mcxjc5@nottingham.ac.uk)
   Ladda Mo-suwan (laddamosuwan@gmail.com)
   Suntharee Kraiveradechachai (suntreekrai@hotmail.com)
   Umaporn Udomsubpayakul (rauub@mahidol.ac.th)
   Chanpen Choprapawon (cchopra2004@gmail.com)

Version: 7 Date: 18 May 2009

Author's response to reviews: see over
18 May 2009

The BioMed Central Editorial Production Team

BioMed Central Ltd
Floor 6, 236 Gray’s Inn Road
London, WC1X 8HL
United Kingdom

Ref: 604686352091131

“Television viewing in Thai infants and toddlers: impacts to language development and parental perceptions”

Dear BioMed Central Editorial Production Team,

We have made the changes in our manuscript as following:

1. Change the spelling of the author, Suntree Kriweradechachai.
2. Change LM’s Email address to ladda.m@psu.ac.th.
3. The word “meant” was replaced by “implied” on page 7 in the measurements section.

We are grateful to you for providing us with an opportunity to publish our manuscript in BMC Pediatrics.

Yours sincerely,

Jariya Chuthapisith

Department of Pediatrics
Faculty of Medicine Ramathibodi Hospital
Mahidol University
Bangkok 10400 Thailand